Author's response to reviews

Title: Borderline gestational diabetes mellitus and pregnancy outcomes

Authors:

Hong Ju (hong.ju@adelaide.edu.au)
Alice R Rumbold (alice.rumbold@menzies.edu.au)
Kristyn J Willson (kristyn.willson@adelaide.edu.au)
Caroline A Crowther (caroline.crowther@adelaide.edu.au)

Version: 4 Date: 29 July 2008

Author's response to reviews: see over
29 July 2008

Dear Editor of BMC Pregnancy and Childbirth,

RE: MS: 1925629563735532

**Borderline gestational diabetes mellitus and pregnancy outcomes**
Hong Ju, Alice R Rumbold, Kristyn J Willson and Caroline A Crowther

Thankyou for your email of 28 July providing confirmation of publication of our manuscript providing we complete formatting changes

We have now made the appropriate revisions and look forward to publication of our manuscript.

Wish best wishes
Yours sincerely,

Professor Caroline Crowther

---

**Formatting changes requested**

1. **Authors' contributions - please use initials for all authors mentioned in this section**

We have now corrected the authors’ contributions section so that initials are used for all authors mentioned.

2. **References - the reference list should contain all authors names, if there are 30 or below. The term et al. should not be used.**

We have now corrected the reference list so that the term et al. is not used and all author names are listed.